Author's response to reviews

Title: Differences in the quality of interpersonal care in complementary and conventional medicine

Authors:

André Busato (andre.busato@memcenter.unibe.ch)
Beat Künzi (beat.kuenzi@swisspep.ch)

Version: 7 Date: 2 November 2010

Author's response to reviews:

Dear Sirs

we modified the manuscript according to the requirements of the editorial production team:

We could not find a double citation of table 2 but the term "2*2 table" was replaced by "two by two table" (line 27 page 10)

The citation of table 3 is on line 8, page 13 and table 5 on line 9, page 14

We completed the citation of figures 2-7 with figure 6 (line 3 page 15)

With best regards

A. Busato